Operability Report 

9/27/2006
Accelerator
Recovery and ORFP of 1 pass 687 MeV 3 way split (325 MeV per linac)
· High current injector setup

· Injector matching

· B Team is looking at lock BPMs

· Waiting for documentation for the new pathlength system

· BCM and ion chamber calibrations

· No CW to BSY dump (Broken pump)

Hall A

Low Q Elastic Scattering
· Up to 20 uA

· Chiller replacement for B Cal

Hall B

GlueEX
· 1 to 30 nA

· Harp PMTs not holding voltage (Fixed)
· 2C24 nA BPM is now working
Hall C

G0 Backwards Angle







· Up to 60 uA

· Weekly Mott and Moeller
· RF phase vs halo
